Author’s response to reviews

Title: Clinical features and genetic analysis of two siblings with Startle disease in an Italian family: a case report.

Authors:

Teresa Sprovieri (teresa.sprovieri@cnr.it)
Carmine Ungaro (carmine.ungaro@cnr.it)
Serena Sivo (s.sivo@hotmail.it)
Michela Quintiliani (michela.quintiliani@yahoo.it)
Ilaria Contaldo (ilaria.contaldo@policlinicogemelli.it)
Chiara Veredice (chiara.veredice@policlinicogemelli.it)
Luigi Citrigno (luigi.citrigno@cnr.it)
Maria Muglia (maria.muglia@cnr.it)
Francesca Cavalcanti (francesca.cavalcanti@cnr.it)
Sebastiano Cavallaro (sebastiano.cavallaro@cnr.it)
Eugenio Mercuri (eugeniomaria.mercuri@unicatt.it)
Domenica Battaglia (domenicaimmacolata.battaglia@unicatt.it)

Version: 2 Date: 28 Feb 2019

Author’s response to reviews:

February 28th, 2019

Dear Prof. Matteo Pasini,

Editor-in-Chief

As required, we have made the corrections:
1. We amended the Consent for publication section and we clarified that all the participants are older than 18. All of them provided a written informed consent.

2. We clarified in the Ethics approval and consent to participate section that all the participants are older than 18. All of them provided a written informed consent.

3. We specified in the Authors’ contributions section the individual contribution of Sebastiano Cavallaro (SC).

4. We included the email addresses for all authors in the title page.

We hope you’ll find our corrections suitable.

Sincerely yours,

Dr. Carmine Ungaro, PhD
Institute of Neurological Sciences, National Research Council
Loc. Burga, 87050 Mangone (CS), Italy
Telephone: +39 0984 98011
Fax: +39 0984 969306
E-mail: carmine.ungaro@cnr.it